Author's response to reviews

Title: Socioeconomic deprivation and environmental noise exposure in Montreal, Canada

Authors:

Laura M Dale (laura.dale@mail.mcgill.ca)
Sophie Goudreau (sgoudrea@santepub-mtl.qc.ca)
Stéphane Perron (sperron@santepub-mtl.qc.ca)
Marianne Hatzopoulou (marianne.hatzopoulou@mcgill.ca)
Audrey Smargiassi (audrey.smargiassi@umontreal.ca)

Version: 3 Date: 1 October 2014

Author's response to reviews:

Dear Editor,

Please find enclosed an original article entitled “Socioeconomic deprivation and environmental noise exposure in Montreal, Canada” to be considered as a research article in Journal of BMC Public Health.

The topic of our article corresponds to the priority of the journal in that we determine whether there is a correlation between socially and materially deprived populations in Montreal and greater exposure to environmental noise to orient Public Health interventions.

This manuscript is original and has not been submitted or published elsewhere in whole and in part.

This manuscript was reviewed by all the authors who agree that the work is ready for submission. All authors thus accept responsibility for the content of this manuscript.

The work presented in this manuscript was supported financially by institutions with no competing interest. Furthermore, none of the authors have competing financial interests regarding this manuscript.

If this manuscript is accepted for publication in BMC Public Health, we give permission to the publisher to reproduce the material found in the article.

Yours sincerely,
Audrey Smargiassi (corresponding author)
audrey.smargiassi@umontreal.ca